Author’s response to reviews

Title: Reducing GHG emissions while improving diet quality: exploring the potential of reduced meat, cheese and alcoholic and soft drinks consumption at specific moments during the day

Authors:
Mirjam van de Kamp (mirjam.van.de.kamp@rivm.nl)
Marije Seves (marijeseves@posteo.nl)
Elisabeth Temme (Liesbeth.Temme@rivm.nl)

Version: 4 Date: 17 Jan 2018

Author’s response to reviews:

Response to editor comment

(1) We note that you have not stated within your 'Ethics approval and consent to participate' section whether you were provided access to the data from the Dutch National Food Consumption Survey following request to the National Institute for Public Health and the Environment. Furthermore, you have not stated here whether you were granted permission, and by who, to use the data from Life Cycle Assessments. We require that you please state any permissions to use the data within this section, and you may remove the information currently provided in line 523-526 of the manuscript.

> The changes as requested by the editor have been made. Please see line 512-522 of the revised manuscript.